Author’s response to reviews

Title: Salivary microflora and mode of delivery: a prospective case control study

Authors:

Katarina Boustedt (katarina.boustedt@regionhalland.se) (katarina.boustedt@live.se)
Josefine Roswall (josefine.roswall@regionhalland.se)
Gunnar Dahlén (gunnar.dahlen@odontologi.gu.se)
Jovanna Dahlgren (jovanna.dahlgren@gu.se)
Svante Twetman (stwe@sund.ku.dk)

Version: 2 Date: 03 Nov 2015

Author’s response to reviews:

Editorial comments:

1. Abstract

Please consider rephrasing the first sentence of subsection "Methods" in the abstract, as the current formulation conveys the impression that the infants themselves volunteered and provided informed consent. Please do not forget to update also the abstract that is separately uploaded in the submission system.

2. Ethics approval

Please confirm in your manuscript that the Regional Ethics Committee in Lund has the authority to approve studies at Halland Hospital.

3. Section titles

Please change section title "Materials and Methods" to "Methods".

Dear editor!

All the above mentioned changes have been made.
Sincerely Katarina Boustedt